Author's response to reviews

Title: An unusual case of orbito-frontal rod fence stab injury with a good outcome.

Authors:

Massimo Miscusi (m.miscusi@libero.it)
Luca De Martino (luca.demartino@libero.it)
Paolo Arangio (paolo.arangio@libero.it)
Fabio De-Giorgio (fabio.degiorgio@rm.unicatt.it)
Piero Cascone (piero.cascone@libero.it)
Antonino Raco (a.raco@libero.it)

Version: 3 Date: 28 April 2013

Author's response to reviews: see over
Dear Editor,

Enclosed please find the revised version of our manuscript entitled “An unusual case of orbito-frontal rod fence stab injury with a good outcome” to be reconsidered for publication in the *BMC Surgery*.

We include a detailed list of all changes made to the manuscript, keyed to the reviewer comments. We hope that after these modifications our manuscript could be suitable for publication in the *BMC Surgery*.

Thank you for all your trouble.

Yours truly,

Riccardo Rossi, M.D., Ph.D. and co-authors
Institute of Legal Medicine, Catholic University, L.go F. Vito 1, 00168 Rome Italy.
Phone: +390635507031. Fax: +390635507033. E-mail: riccardo.rossi@rm.unicatt.it
We welcome reviewers’ comments and suggestions and have modified the manuscript according to their advice. Please find our itemized, point-by-point response to the comments of the reviewers below with the indication of the changes we made.

ANSWERS TO REVIEWER COMMENTS:

Reviewer: 3
*Only two sentences left inattentively in the revised should be removed:*
  
  Background, page 1, line 7: “Unlike missile injuries, cranial stab wounds…”
  Conclusions, page 4, line 8: “anche qui…” (sentence in Italian language)

The two sentences have been removed.